\begin{table}%T1
\caption{\vinef: Some basic parameters.}
\label{tabgalpar}
%\centerline
 {
\begin{tabular}{lcc} \hline\hline
 & \vinef  & Reference$^a$ \\
\hline
Alternate name & IRAS 09293+2143 & \\
Type & SAB(rs)bc & 1\\
RA (2000) & 09$^{\rm h}$32$^{\rm m}$09\fs9 & 2\\
Dec (2000) & +21\degr 30\arcmin 07\arcsec & 2\\
Distance/Mpc & 8.9  & 3\\
Radial velocity/\kms & 556 & 3\\
$D_{25}$ diameter & 11\farcm6 & 1 \\
Axial ratio & 0.49 & 1\\
Inclination & 66\degr & 1 \\
log ($L_{\rm B}$ /\Lsun) & 10.13 & 1 \\
log ($M_{\rm HI}$/\Msun) & 9.25 & 1 \\
log ($N_{\rm HI}$/cm$^{-2}$) (Galactic) & 20.45 & 4 \\
log ($N_{\rm HI}$/cm$^{-2}$) (intrinsic) & 20.92 & 5 \\
\hline
\end{tabular}}
\smallskip
$^a$ References:\\
\hspace*{2mm}
1) \cite{tul88x}.\\
\hspace*{2mm}
2) \cite{fal99x}.\\
\hspace*{2mm}
3) \cite{dro00x}.\\
\hspace*{2mm}
4) \cite{dic90x}.\\
\hspace*{2mm}
5) \cite{wev86x}.
\end{table}